Washington University School of Medicine Digital Commons@Becker

Open Access Publications

2018

De novo variants in congenital diaphragmatic hernia identify MYRF as a new syndrome and reveal genetic overlaps with other developmental disorders

Brad W. Warner Washington University School of Medicine in St. Louis

et al.

Follow this and additional works at: https://digitalcommons.wustl.edu/open_access_pubs

Recommended Citation

Warner, Brad W. and et al., "De novo variants in congenital diaphragmatic hernia identify MYRF as a new syndrome and reveal genetic overlaps with other developmental disorders." PLoS Genetics. 14,12. e1007822. (2018). https://digitalcommons.wustl.edu/open_access_pubs/7731

This Open Access Publication is brought to you for free and open access by Digital Commons@Becker. It has been

accepted for inclusion in Open Access Publications by an authorized administrator of Digital Commons@Becker. For more information, please contact vanam@wustl.edu.



G OPEN ACCESS

Citation: Qi H, Yu L, Zhou X, Wynn J, Zhao H, Guo Y, et al. (2018) *De novo* variants in congenital diaphragmatic hernia identify *MYRF* as a new syndrome and reveal genetic overlaps with other developmental disorders. PLoS Genet 14(12): e1007822. https://doi.org/10.1371/journal.pgen.1007822

Editor: Stefan Mundlos, Max Planck Institute for Molecular Genetics, GERMANY

Received: June 13, 2018

Accepted: November 8, 2018

Published: December 10, 2018

Copyright: © 2018 Qi et al. This is an open access article distributed under the terms of the <u>Creative</u> <u>Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Whole genome sequencing data can be obtained from dbGAP through accession phs001110.

Funding: Some exome sequencing was provided by the University of Washington Center for Mendelian Genomics (UW-CMG) and was funded by the National Human Genome Research Institute and the National Heart, Lung and Blood Institute grant HG006493 to DN, MB, and Suzanne Leal. The whole genome sequencing data were generated RESEARCH ARTICLE

De novo variants in congenital diaphragmatic hernia identify *MYRF* as a new syndrome and reveal genetic overlaps with other developmental disorders

Hongjian Qi^{1,2}, Lan Yu³, Xueya Zhou^{1,3}, Julia Wynn³, Haoquan Zhao^{1,4}, Yicheng Guo¹, Na Zhu^{1,3}, Alexander Kitaygorodsky^{1,4}, Rebecca Hernan³, Gudrun Aspelund⁵, Foong-Yen Lim⁶, Timothy Crombleholme⁶, Robert Cusick⁷, Kenneth Azarow⁸, Melissa E. Danko⁹, Dai Chung⁹, Brad W. Warner¹⁰, George B. Mychaliska¹¹, Douglas Potoka¹², Amy J. Wagner¹³, Mahmoud ElFiky¹⁴, Jay M. Wilson^{15,16}, Debbie Nickerson¹⁷, Michael Bamshad¹⁷, Frances A. High^{15,16,18}, Mauro Longoni^{16,18}, Patricia K. Donahoe^{16,18}, Wendy K. Chung^{3,19,20*}, Yufeng Shen^{1,4,21*}

1 Department of Systems Biology, Columbia University Medical Center, New York, New York, United States of America, 2 Department of Applied Mathematics and Applied Physics, Columbia University, New York, New York, United States of America, 3 Department of Pediatrics Medical Center, Columbia University, New York, New York, United States of America, 4 Department of Biomedical Informatics, Columbia University Medical Center, New York, New York, United States of America, 5 Department of Surgery, Columbia University Medical Center, New York, New York, United States of America, 6 Cincinnati Children's Hospital, Cincinnati, Ohio, United States of America, 7 Children's Hospital & Medical Center of Omaha, University of Nebraska College of Medicine, Omaha, Nebraska, United States of America, 8 Department of Surgery, Oregon Health & Science University, Portland, Oregon, United States of America, 9 Monroe Carell Jr. Children's Hospital, Vanderbilt University Medical Center, Nashville, Tennessee, United States of America, 10 Washington University, St. Louis Children's Hospital, St. Louis, Missouri, United States of America, 11 University of Michigan, CS Mott Children's Hospital, Ann Arbor, Michigan, United States of America, 12 Children's Hospital of Pittsburgh, Pittsburgh, Pennsylvania, United States of America, 13 Medical College of Wisconsin, Milwaukee, Wisconsin, United States of America, 14 Department of Pediatric Surgery, Faculty of Medicine, Cairo University, Cairo, Egypt, 15 Department of Surgery, Boston Children's Hospital, Boston, Massachusetts, United States of America, 16 Department of Surgery, Harvard Medical School, Boston, Massachusetts, United States of America, 17 University of Washington, Seattle, Washington, United States of America, 18 Pediatric Surgical Research Laboratories, Department of Surgery, Massachusetts General Hospital, Boston, Massachusetts, United States of America, 19 Department of Medicine, Columbia University, New York, New York, United States of America, 20 Herbert Irving Comprehensive Cancer Center, Columbia University Medical Center, New York, New York, United States of America, 21 JP Sulzberger Columbia Genome Center, Columbia University Medical Center, New York, New York, United States of America

So These authors contributed equally to this work.

* wkc15@columbia.edu (WKC); ys2411@cumc.columbia.edu (YS)

Abstract

Congenital diaphragmatic hernia (CDH) is a severe birth defect that is often accompanied by other congenital anomalies. Previous exome sequencing studies for CDH have supported a role of *de novo* damaging variants but did not identify any recurrently mutated genes. To investigate further the genetics of CDH, we analyzed *de novo* coding variants in 362 proband-parent trios including 271 new trios reported in this study. We identified four unrelated individuals with damaging *de novo* variants in *MYRF* (P = 5.3×10^{-8}), including one likely gene-disrupting (LGD) and three deleterious missense (D-mis) variants. Eight



through NIH Gabriella Miller Kids First Pediatric Research Program (X01HL132366 and X01HL136998). This work was supported by NIH grants: National Institute of Child Health and Human Development R01HD057036 (LY, JMW, WKC), National Heart, Lung, and Blood Institute R03HL138352 (AK, WKC, YS), National Institute of General Medical Sciences R01GM120609 (HQ, YS), National Center for Research Resources UL1 RR024156 (WKC), and National Institute of Child Health and Human Development 1P01HD068250 (PKD, ML, FAH, JMW, WKC, YS). Additional funding support was provided by grant from CHERUBS, a grant from the National Greek Orthodox Ladies Philoptochos Society, Inc. and generous donations from Wheeler Foundation, Vanech Family Foundation, Larsen Family, Wilke Family and many other families. Funders did not play any role in study design, data collection, analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

additional individuals with *de novo* LGD or missense variants were identified from our other genetic studies or from the literature. Common phenotypes of *MYRF de novo* variant carriers include CDH, congenital heart disease and genitourinary abnormalities, suggesting that it represents a novel syndrome. *MYRF* is a membrane associated transcriptional factor highly expressed in developing diaphragm and is depleted of LGD variants in the general population. All *de novo* missense variants aggregated in two functional protein domains. Analyzing the transcriptome of patient-derived diaphragm fibroblast cells suggest that disease associated variants abolish the transcription factor activity. Furthermore, we showed that the remaining genes with damaging variants in CDH significantly overlap with genes implicated in other developmental disorders. Gene expression patterns and patient phenotypes support pleiotropic effects of damaging variants in these genes on CDH and other developmental disorders. Finally, functional enrichment analysis implicates the disruption of regulation of gene expression, kinase activities, intra-cellular signaling, and cytoskeleton organization as pathogenic mechanisms in CDH.

Author summary

Congenital diaphragmatic hernia (CDH) is a life-threatening condition affecting about 1 every 3000 newborns. Although the role of genetics in the pathogenesis of CDH has been well established, only a handful of disease genes have been identified so far. We and other have previously shown that *de novo* variants, those carried by the cases but not inherited from parents, are enriched in sporadic CDH cases consistent with their negative effects on reproductive fitness. To further investigate the genetics of CDH, we analyzed de novo variants in 362 proband-father-mother trios from whole exome or genome sequencing data and identified four patients carrying damaging variants in MYRF, a membrane associated transcription factor that is highly expressed in developing diaphragm and heart. We then ascertained a total of 12 patients with MYRF de novo variants, and found they shared common phenotype characteristics including congenital abnormalities in diaphragm, heart and reproductive organs. The high rate of recurrence and similar phenotypic manifestations suggest that *de novo* variants of *MYRF* have pleiotropic effects and cause a novel syndrome. The identified new gene is reminiscent of previously identified CDH genes (e.g., GATA4, GATA6, NR2F2, ZFPM2, and WT1) that are also associated with other developmental disorders. Indeed, we found in our cohort more than 20 damaging de novo variants in genes implicated in other developmental disorders but not previously linked to CDH. The overlap was unlikely to occur by chance and can be best explained by their pleiotropic effects. We also showed that, despite the shared genetic basis with other disorders, damaging de novo variants in CDH as a whole were enriched in specific functional pathways that recapitulated our current knowledge about diaphragm development. So additional candidate genes can be prioritized based on the genetic pleiotropy and functional specificity. The findings have general implications in design and analysis in genetic studies of rare birth defects.

Introduction

Congenital diaphragmatic hernia (CDH) is a severe developmental disorder affecting 1 in 3000 live births $[\underline{1}, \underline{2}]$. It is characterized by defects in diaphragm that allow the abdominal

viscera to move into the thoracic cavity and is associated with pulmonary hypoplasia and in some cases pulmonary hypertension. CDH can be isolated (50–60%) or associated with anomalies in other organs including the heart, brain, kidneys and genitalia [3, 4]. Despite advances in treatment, mortality rate remains high [5, 6]. A better understanding of the causative factors for CDH may inform disease prevention and treatment.

The genetic contribution to CDH has been established by familial aggregation [7], rare monogenic disorders associated with CDH in humans [8], chromosome abnormalities [9], copy number variations [10–12], and mouse models [13]. However, our understanding of the genetic basis of CDH is still rudimentary. The historically low reproductive fitness of individuals with CDH led to the hypothesis that *de novo* variants with large effect sizes may explain a fraction of CDH patients as in other developmental disorders [14, 15]. We and others have previously reported an enrichment of damaging variants in sporadic CDH patients [16, 17]. However, no recurrently mutated gene was identified in our genome wide analyses due to the limited sample size.

To continue the search for new CDH genes, we performed whole exome (WES) or whole genome sequencing (WGS) of 271 new trios. Combined with previously published WES data [16, 17], we analyzed all 362 trios. We confirmed the overall burden of damaging *de novo* variants and identified a new disease gene recurrently mutated in cases with similar syndromic features. To prioritize additional risk genes, we analyzed cross-disorder overlap and pathway enrichment. The results provide insights into the genetic architecture of CDH and suggest additional candidate genes.

Results

Sample characteristics

Patients were recruited from the multicenter, longitudinal DHREAMS (Diaphragmatic Hernia Research & Exploration; Advancing Molecular Science) study [11]. We excluded patients with known genetic causes from clinical karyotype or chromosome microarray or with a family history of CDH. WES was performed on 118 proband-parents trios, a subset (39) of whom were published previously [17]. WGS was performed on 192 trios including 27 without damaging variants from the previous study [17]. On average, 91% of coding regions in WES samples and 98% in WGS samples were covered by 10 or more unique reads (S1 Fig). WGS showed more uniform distribution of sequencing depth that contributes to higher power in detecting coding variants [18, 19]. For the 27 overlapping samples, 12 additional *de novo* coding variants were identified in WGS including 10 not included in the exome targets or with low depth of coverage and two that failed stringent QC filters in our previous study.

Combined with trios collected by Boston Children's Hospital/Massachusetts General Hospital (BCH/MGH) [16], we analyzed a total 362 unique trios (S1 Table). Clinical and demographic information of patients are given in S1 Data. In the combined cohort, there were 212 (58.6%) male and 150 (41.4%) female patients. The male-to-female ratio (1.4:1) was consistent with published retrospective and prospective cohorts [20, 21]. The most common type of CDH was left-sided Bochdalek; rare forms of CDH or atypical lesion sides were also included (Table 1).

A total 149 (41.2%) cases had additional congenital anomalies or neurodevelopmental disorders (NDD) at the time of last follow up and were classified as complex cases; and 209 (57.7%) patients had no additional anomalies at last contact were classified as isolated cases. The most frequent comorbidity among complex cases was cardiovascular anomalies (44.3%). NDD, skeletal malformations, and genitourinary defects were also observed in complex cases (Table 1).

	Number	Percent
Gender		
Male	212	58.6%
Female	150	41.4%
CDH classification		
Isolated	208	57.5%
Complex	149	41.2%
Unknown	5	1.4%
Lesion side		
Left	270	74.6%
Right	56	15.5%
Eventration/Morgagni/Agenesis	11	3.0%
Unknown	25	6.9%
CDH type		
Bochdaleck	294	81.2%
Other [#]	22	6.1%
Unknown	46	12.7%
DHREAMS cohort (n = 283): Time of recruitme	nt	
Neonatal	229	80.9%
Fetal	9	3.2%
Child	45	15.9%
Discharge vital status (n = 283)		
Survived	241	85.2%
Deceased	42	14.8%
Development assessment ⁹ (n = 283)		
At 2 years follow-up	152	53.7%
At 5 years follow-up	70	24.7%
No assessment at either 2 or 5 years	128	45.2%
Additional anomalies in complex cases (n = 149)		
Cardiovascular	66	44.3%
Neurodevelopmental [§]	37	24.8%
Skeletal	26	17.4%
Genitourinary	14	9.4%
Gastrointestinal	13	8.7%

Table 1. Clinical summary of patients.

9 Development assessment at 2 years follow-up include Vineland Adaptive Behavior Assessment (Vineland-II) and/ or Bayley Scales of Toddler Development (Bayley-III); tests at 5 years follow-up include Vineland-II and/or Wechsler Preschool and Primary Scale of Intelligence (WPPSI).

\$Neurodevelopmental conditions include congenital abnormalities in central nerves system, and developmental delay or neuropsychiatric disorders based on the follow-up developmental evaluations.

https://doi.org/10.1371/journal.pgen.1007822.t001

Burden of de novo coding variants

We identified 471 coding *de novo* variants in 264 (72.9%) cases including 430 single nucleotide variants (SNV) and 41 indels. Transition-to-transversion ratio of *de novo* SNVs was 2.64. The number of *de novo* coding variants per proband closely followed a Poisson distribution, with an average of 1.32 in WGS trios and 1.28 in combined WES trios (S2 Fig). Variants that were likely gene disrupting (LGD) or predicted deleterious missense ("D-mis" defined by CADD score [22] \geq 25) were considered as damaging. A total of 193 damaging variants (57 LGD and

138 D-mis) were identified in 150 (41.4%) cases, including 38 (10.5%) cases harboring two or more such variants. Compared with the baseline expectations (Material and methods) [23], both *de novo* LGD variants (0.16 per case) and D-mis variants (0.38 per case) were significantly enriched in cases (fold enrichment (FE) = 1.73, P = 8.6×10^{-5} by one-sided Poisson test for LGD; FE = 1.5, P = 1.1×10^{-6} for D-mis) while the frequency of silent variants closely matched the expectation (0.30 per case, FE = 1.01, P = 0.48 by one-sided Poisson test).

Consistent with the previous study [16], damaging variants showed a higher enrichment in complex cases than isolated cases (FE = 1.70 vs 1.64 for LGD, 1.61 vs 1.38 for D-mis; <u>S2 Table</u>); and the proportion of complex cases who carried damaging variants was higher than isolated cases (43.6% vs. 39.4%). Burden of damaging variants was also higher in female than male cases (FE = 2.09 vs 1.47 for LGD, 1.63 vs 1.36 for D-mis; <u>S2 Table</u>), supporting a "female protective model" similar to autism and other NDD with male bias [24, 25].

Recent studies highlighting the use of large population reference sequencing data in interpreting LGD variants has demonstrated that genes depleted of LGD variants in the general population were more likely associated with disorders with reduced reproductive fitness[26]. We defined constrained genes by the estimated probability of loss-of-function intolerance (pLI) [27] \geq 0.5 and found the burden of LGD variants was largely explained constrained genes (Table 2). D-mis also showed a higher enrichment in constrained genes (Table 2).

MYRF is a new syndromic CDH gene

We identified eight genes affected by more than one *de novo* LGD or missense variant (<u>S3</u> <u>Table</u>). The top ranked gene, *MYRF*, has one frameshift insertion and three damaging missense variants, all of which were validated by Sanger sequencing. It is the only constrained gene in the list. By comparing with baseline expectations, only *MYRF* reaches genome-wide significance after Bonferroni correction of ~20000 coding genes (P = $5.3 \times 10^{-8} < 0.01/20000$, by one-sided Poisson test).

Notably, all four patients with *MYRF* variants also had congenital heart disease (CHD), and three of them had genital anomalies including blind-ending vagina in a female and ambiguous genitalia or undescended testes in two male cases (<u>Table 3</u>). By screening another 220 CDH

Gene Sets	Variant class	Number of variants	Baseline expectation	Fold enrichment	P-value
All Genes	Synonymous	110	109.1	1.01	0.48
	Missense	295	250.6	1.18	3.42E-03
	D-mis	138	93.7	1.47	1.08E-05
	LGD	57	32.9	1.73	8.60E-05
Constrained Genes	Synonymous	34	38.8	0.88	0.80
	Missense	112	88.1	1.27	7.91E-03
	D-mis	59	38.0	1.55	9.39E-04
	LGD	30	12.0	2.50	9.05E-06
Other Genes	Synonymous	76	70.3	1.08	0.26
	Missense	184	162.6	1.13	0.053
	D-mis	80	55.7	1.44	1.28E-03
	LGD	27	20.9	1.29	0.11

Table 2.	Burden	of de	novo	coding	variants
----------	--------	-------	------	--------	----------

Constrained genes are defined by pLI metrics \geq 0.5. LGD: likely gene disrupting, including frameshift, stop-gain, stop-loss, and variants at canonical splice sites; D-mis: predicted deleterious missense variants defined by CADD Phred score \geq 25. The baseline expectations for different types of variants were calculated by the previous published method[23, 28]. The enrichment of observed number of variants was evaluated by a one-sided Poisson test.

https://doi.org/10.1371/journal.pgen.1007822.t002

Study	Sample ID	Genetic Sex [#]	<i>De novo</i> variant (NM_001127392.2)	CADD Phred	Diaphragm defect	Cardiovascular defect	Urogenital defect	Other malformations
Current study	01-1008	XY	c.235dupG: p.G81Wfs*45	-	L-CDH	ASD,VSD,ToF	Bilateral undescended testes	No
	01-0429	XX	c.1303G>A:p.G435R	32	L-CDH	VSD	No internal genital organs, blind-ending vagina	Accessory spleen
	04-0042	XY	c.2036T>C:p.V679A	25.9	L-CDH	ASD,VSD	Unknown	Unknown (Deceased)
	05-0050	XY	c.2084G>A:p.R695H	34	CDH	HLHS	Ambiguous genitalia, undescended testes	Intellectual disability and motor delay at 2 years old
	01-0033	XX	c.1904-1G>A	25	R-CDH	Scimitar syndrome	Unknown	Unknown (Deceased)
	01- 0591*	XX			No	HLHS	Unknown	Unknown (Deceased)
	CHU-11	XY	c.1786C>T:p.Q596*	37	No	Dextrocardia	Swyer syndrome with female genitalia	Right pulmonary hypoplasia
PCGC [29]	1-02264	XY	c.1160T>C:p.F387S	27.9	No	AAH, CoA, HLHS	Ambiguous genitalia, hypospadias, undescended testis	No
	1-03160	XY	c.1209G>C:p.Q403H	27.6	Right hemi- diaphragm eventration	Scimitar syndrome, AAH, ASD, BAV, HLHS, MS, VSD	Undescended testis	Lung hypoplasia
	1-07403	XY	c.1435C>G:p.L479V	23.9	No	BAV, CoA	Swyer syndrome with female genitalia	Short stature
Pinz et al. [<u>30</u>]	Case 1	XY	c.2336+1G>A	26.8	No	Scimitar syndrome, cor triatriatum	Penoscrotal hypospadias, micropenis, unilateral cryptorchidism	Mild speech delay, pulmonary hypoplasia, tracheal anomalies
	Case 2	XY	c.2518C>T:p.R840*	44	R-CDH	Scimitar syndrome	Persistent urachus, Undescended testis	Cleft spleen, thymic involution, thyroid fibrosis
Chitayat et al. [<u>31</u>]	Fetus case	XY	c.1254_1255dupGA: p. T419RfsX14	-	No	HLHS	Ambiguous external genitalia, right hepato- testicular fusion and left spleno-testicular fusion	Mild pulmonary hypoplasia, intestinal malrotation

Table 3. Phenotype characteristics of patients with *de novo* coding variants in MYRF.

Abbreviations: L/R-CDH, (lef/right)-congenital diaphragmatic hernia; AAH, aortic arch hypoplasia; ASD, atrial septal defect; BAV, bicommissural aortic valve; CoA, coarctation of aorta; VSD, ventricular septal defects; ToF, Tetralogy of Fallot; MS, mitral stenosis; PCGC, Pediatric Cardiovascular Genetics Consortium. * 01–0591 is the monozygotic twin of 01–0033.

Genetic sex is based upon the chromosome complement.

https://doi.org/10.1371/journal.pgen.1007822.t003

trios collected by the DHREAMS study, we identified another patient harboring a *de novo* splice acceptor site variant. The female patient had a diagnosis of Scimitar syndrome (a complex form CHD). She also had a monozygotic twin sister with hypoplastic left heart syndrome who also carried the same variant but no known CDH.

Given the strong association of *MYRF* variants with CHD, we then searched for *de novo* variants from a recently published study of CHD conducted by Pediatric Cardiac Genomics Consortium (PCGC) [29] and identified three additional *de novo* missense variants in *MYRF* from 2645 trios. All CHD patients also had genitourinary anomalies, including a patient with Swyer syndrome (46XY karyotype with female reproductive organs). One CHD patient with the Q403H variant had hemidiaphragm eventration. Recently, Pinz *et al.* [30] and Chitayat et al [31] reported three additional cases with complex CHD who carried *de novo* LGD variants in *MYRF*. All cases had genital defects, and one had CDH and the other two had pulmonary hypoplasia. Furthermore, from clinical WES, we also identified a Swyer syndrome patient with a stop-gain variant in *MYRF* who had dextrocardia and pulmonary hypoplasia.



Fig 1. *De novo* coding variants in MYRF and their functional impact on transcriptome. (a) Schematic diagram of the MYRF protein structure. DBD: DNA binding domain; ICA: Intramolecular chaperone auto-processing domain; Pro-rich: proline-rich region; TM: transmembrane helix. The position of DBD and ICA were based on the annotation from InterPro, and Pro-Rich and TM were from SwissProt. The coordinates are given with respect to the canonical isoform (1151 amino acids). The relative position of 12 *de novo* ocding variants are displayed, including 6 discovered in the current study (shown in red), and five from published studies of congenital heart disease (CHD) [29, 30] (shown in blue). LGD variants are shown on top of the protein; and missense variants are on the other side. Shown below the protein structure is the density of missense variants in gnomAD (http://gnomad.broadinstitute.org/). A missense constraint region [37] is highlighted in red (observed/expected number of missense variants = 0.31) (b) Z-score for each gene is the standardized expression level across samples. Mean Z-scores of MYRF target genes in three *MYRF* variant carriers were shifted to the lower end as compared with other genes. (c) Gene-set enrichment analysis (GESA) was applied to genes ranked by the estimated fold change of expression level comparing *MYRF* variant carriers with other cases. The MYRF target genes tend to have lower ranks and majority of them were down-regulated in *MYRF* variant carriers (NES = -2.10, P<5.0E-4).

https://doi.org/10.1371/journal.pgen.1007822.g001

PLOS GENETICS

In total, we identified 13 patients harboring 12 different *de novo* functional variants in *MYRF* (6 LGD and 6 missense variants; Fig 1A). All patients had CHD; and excluding those who died in infancy and had incomplete phenotypic information, all patients also had genito-urinary anomalies. CDH was present in 7 out of 12 patients, and diaphragm defects were not systematically evaluated in cases without reported CDH. There was no clear phenotypic difference between patients with LGD variants and those with missense variants (Table 3). Taken

together, the unique association of CDH and similar non-diaphragm defects including CHD, Scimitar syndrome, genitourinal anomalies and sex reversal in 46XY patients with *de novo* variants in *MYRF* establish it as a new syndromic CDH gene.

MYRF is a highly constrained gene in the population (pLI = 1). By examining both public databases (ExAC and gnomAD) and our own cohort, we only identified two rare LGD variants that affect all functional isoforms, yet their functional consequences were not clear (S5 Table). We also searched for inherited variants in 362 CDH trios and 2645 CHD trios from PCGC but did not find any inherited LGD variants in probands. Enrichment for *de novo* LGD variants associated with CDH and near complete absence of loss-of-function variants in the general population suggest that variants causing loss of *MYRF* function are likely fully penetrant for one or more aspects of this syndrome. All six *de novo* missense variants identified patients were also absent from the public databases, consistent with their high penetrance as LGD variants in this gene.

Functional analysis of MYRF variants

MYRF is a membrane-associated transcription factor that plays a pivotal role in oligodendrocyte differentiation and myelination [32, 33]. Although it has not previously been implicated in diaphragm or cardiac development, its expression level was ranked at the top 21% of genes expressed in mouse developing diaphragm at E11.5 [34] and top 14% in developing heart at E14.5 [35].

The MYRF protein has two functional isoforms. Both isoforms contain a N-terminal proline-rich region followed by a DNA binding domain (DBD), which can be cleaved from the membrane by a region called intramolecular chaperon auto-processing (ICA) domain. All frameshift and stop gained variants resulted in truncated protein products in both functional isoforms and may trigger non-sense mediated decay. The precise functional effects of splice site variants were not evaluated, but are predicted to cause exon skipping, intron retention or activation of cryptic splice site and also result in a truncated protein. All six missense variants aggregated in the two DBD and ICA functional domains (Fig 1a). The missense variants were predicted as deleterious by a majority of bioinformatics tools (S4 Table). Most of the affected amino acid residues are highly conserved across species (S3 Fig).

MYRF DBD is homologous to yeast transcriptional factor Ndt80 but MYRF can only function as a trimer [36]. All missense variants in this domain are located in a region depleted of missense variants in the population (observed/expected = 0.31; Fig 1A) and have high MPC scores [37] (S4 Table). Protein structure modeling predicted that those variants may affect DNA binding affinity (F387S), change surface charge distribution (Q403H), or destabilize the protein structure (G435R and L479R) (S4 Fig).

Previous studies also showed that full length MYRF forms a trimer before cleavage, and trimerization is required for auto-cleavage and subsequent activation [<u>38</u>]. The ICA domain which is distantly related to bacteriophage's tailspike protein was believed to play an essential role in MYRF trimerization. Two missense variants (V679A, R695H) are located at the C-terminal end of the ICA domain where the triplet helix bundle is formed [<u>39</u>]. V679 is one of the critical residues in ICA that is fully conserved from human to bacteriophage (<u>S3 Fig</u>). Structure modeling predicted that the variant R695H may destabilize the trimer structure (<u>S4 Fig</u>) and would fail to produce functional MYRF DBD trimers by trimerization-dependent auto-proteolysis.

To evaluate the effect of *MYRF* variants on gene expression, we performed RNA-seq on diaphragm fibroblast cell cultures from neonatal patients. After removing outlier samples (<u>S5</u> <u>Fig</u>), we obtained transcriptome data of 31 patients including three with a *de novo MYRF* variant (one frameshift insertion and two missense variants in the ICA domain). Most patients (27/31, 87%) included in the RNA-seq analysis were self-reported non-Hispanic White. Additionally, we identified 74 putative MYRF target genes from a previous study of rat oligodendrocyte progenitor cells (<u>S3 Data</u>) [40]. Gene expression levels were quantified as TPM (transcripts per million mapped reads). The z-scores of expression levels of putative MYRF target genes were systematically shifted down in MYRF mutant cells (P = 2.4E-7 by Kolmogorov-Smirnov test; Fig 1B), consistent with the reduced transcription factor activities caused by the damaging variants. We quantified differential expression (DE) of genes between samples with and without *de novo MYRF* variants by a shrinkage estimator of fold change [41]. Selected DE genes were validated by quantitative polymerase chain reaction (qPCR) on the same cell cultures (S7 Fig). Using gene set enrichment analysis [42] of genes ranked by the fold changes, putative MYRF target genes are significantly enriched among the down-regulated genes (normalized enrichment score (NES) = -2.10, P< 5.0E-4; Fig 1C). Since all MYRF mutation carriers were males, we repeated the analysis using only males and found the results are similar as using all samples (NES = -1.95, P<5.0E-4), suggesting that sex is not a confounding factor. The patient with the MYRF frameshift variant was the only MYRF mutation carriers whose ethnicity was not self-reported White. The enrichment of MYRF target genes is also observed in genes down-regulated in the two samples with missense variants (S6 Fig), suggesting that the result was not driven by the LGD variant or ethnicity.

Manual inspection of top DE genes (S4 Data) revealed that *GATA4*, a known CHD gene that has also been implicated in familial and sporadic CDH [43], was significantly down-regulated in cases with *de novo MYRF* variants (estimated fold change = 0.54, q-value = 0.03). Interestingly, we observed that expression trajectories of *MYRF* and *GATA4* were similar in mouse developing diaphragm and lung (S8 Fig) suggesting that they play similar functional roles during diaphragm and pulmonary development.

Besides *MYRF*, we estimated there were 64 (95% CI: 38–93) genes with *de novo* variants implicated in CDH based on the overall burden analysis. Most of those genes have only one damaging variant in the cohort. To prioritize among all the genes with *de novo* damaging variants, we took two approaches.

Genetic overlap with other disorders

We noted that CHD was the most common non-diaphragm defect in complex cases (Table 1). Damaging mutations in *MYRF* have been identified in a previous CHD study but the gene did not reach genome-wide significance [29]. The identification of the *MYRF* syndrome suggested that the comorbidity of CHD and CDH in some cases can be explained by the same genetic factors, many of which remain to be discovered. CDH is also part of the phenotype spectrum of several rare Mendelian disorders [8]. Recently discovered genes for developmental disorders are often pleiotropic and implicated in multiple diseases [15, 29, 44]. Thus, the finding of *MYRF* motivated us to assess the genetic overlap between CDH and other developmental disorders, especially CHD, to help us prioritize additional CDH genes with pleiotropic effects. To this end, we curated genes that were known or implicated in CHD and other developmental disorders (S5 Data; Materials and Methods). Hereafter we refer to these known or candidate genes as CHD or DD genes.

In addition to *MYRF*, we identified a total of 26 DD/CHD genes with damaging *de novo* variants in 25 CDH patients (Fig 2A). Using a simulation approach that accounted for the number of variants, gene size, and sequence context (Materials and Methods), we found that damaging variants in CDH were significantly enriched in the DD and CHD genes (Fig 2B). For example, we observed 6 CHD genes with *de novo* LGD variants in CDH which was 4.7-fold higher than expected ($P = 1.7 \times 10^{-3}$); the number of DD genes with *de novo* LGD



Fig 2. Genetic overlap with other developmental disorders. (a) Venn diagram shows 25 genes implicated in developmental disorders and congenital heart disease (DD and CHD genes, Materials and Methods) that are affected by damaging variants in CDH. (b) Enrichment of LGD and D-mis variants in DD and CHD genes. Enrichment was evaluated by comparing the observed number of *de novo* damaging variants in DD and CHD genes with the expected number of hits by randomly scattering the same number of variants to the exome while controlling for the number of variants, gene length, and sequence context (Materials and Methods). (c) Expression percentile ranks in the developing diaphragm [34], heart [70] and brain [70] are shown for all genes (green density), highlighting DD and CHD genes listed in (b). Smaller ranks correspond to higher expression levels.

https://doi.org/10.1371/journal.pgen.1007822.g002

variants (8) was 3.4 folder higher than expected ($P = 2.3 \times 10^{-3}$). Among CHD genes with at least one damaging variant in CDH, haploinsufficiency of *WT1* is a known cause of several syndromic forms of CDH [8]; *ZFPM2* and *GATA6* have already been established as CDH genes by previous studies [45, 46]. However, the enrichment of damaging variants and especially LGD variants remained significant after excluding known or candidate CDH genes [47] (S9 Fig). Furthermore, the enrichment cannot fully be explained by the over-representation of constrained genes, because the enrichment persisted after conditioning on all constrained genes and remained significant for LGD variants (S9 Fig). The cross-disease overlap suggests that pleiotropic effects of variants in the genes associated with other developmental disorders are also associated with CDH in a fraction of cases. Since CHD genes were curated based on the damaging mutations in CHD patients and DD genes were mostly implicated in other developmental disorders, the genes that appear in both sets were more likely to participate in a broader range of developmental process. Accordingly, the enrichment in genes found exclusively in one set was significantly reduced (Fig 2B, S9 Fig).

We reviewed the most recent medical records of those patients (<u>S7 Table</u>) and identified six complex cases with CHD and/or NDD compatible with the initial reported phenotypes for these genes. Two additional cases were found to have non-CHD cardiovascular defects like two-vessel cord or dilated aortic root; and another four had mild-to-moderate developmental delay/intellectual disability at latest evaluation. Four patients who carried LGD variants in known DD genes (*POGZ*, *ARID1B*, *FOXP1*, and *SIN3A*) and one patient who carried a known activating variant in the Noonan syndrome gene *PTPN11* were considered pathogenic variants by the American College of Medical Genetics and Genomics guidelines [<u>48</u>].

Pleiotropy was further supported by the gene expression data. The majority of the 26 DD/ CHD genes with damaging *de novo* variants in CDH were not only highly expressed in mouse developing diaphragm but also in developing heart or brain (Fig 2C). Indeed, over all coding genes, expression ranks in the three developing organs were highly correlated (Spearman rank correlation r = 0.74 between diaphragm and heart, 0.74 between diaphragm and brain). Therefore, high diaphragm expression can be a proxy for a pleiotropic effect. Consistent with this, we found that all damaging *de novo* variants in complex cases, presumed to enrich causative variants affecting multiple organs, were greatly enriched in genes at the top quartile of expression in developing diaphragm (FE = 4.6, P = 7.9x10⁻⁷ by one-sided Poisson test for LGD; FE = 2.4, P = $1.8x10^{-4}$ for D-mis). By contrast, in isolated cases, the enrichment of damaging variants was distributed in genes across a broad range of expression (Fig 3).

Functional enrichment map

As a second approach to prioritize CDH genes, we hypothesized that different CDH genes converge onto a small number of pathways, and novel genes in the enriched pathways could be candidates for new disease genes. We evaluated functional enrichment of genes affected by damaging *de novo* variants to identify biological processes involved in CDH. To boost the signal, only constrained genes or known haploinsufficient genes were included in the pathway analysis (Materials and Methods). A total of 63 Gene Ontology Biological Process gene sets were enriched at a false discovery rate (FDR) of 0.1 (S6 Data). To remove the redundancies between gene sets, we used a similarity score to organize functionally related gene sets into a network. The resulting network was annotated and visualized as a functional enrichment map (Fig 4A). Eleven functional groups were identified that recapitulated our current knowledge about the molecular genetic basis of CDH [49]. They were supported by 48 genes including 27 novel genes (Fig 4B).

Transcription factor haploinsufficiency is an established cause of CDH [50] and other birth defects [51]. Recently, disruption of epigenetic machinery was also found to underlie many developmental disorders [35, 44, 52]. The majority of DD/CHD genes directly or indirectly regulate gene expression which formed a highly connected cluster of enriched gene sets, some of the transcription factors are involved in the development of heart, lung and reproductive organs. We identified nine novel genes encoding transcription factors or histone modifiers.

Proper cell migration is critical during diaphragm development. Initially, mesenchymal precursor cells migrate from mesoderm to form the primordial diaphragm. After that, pleuro-peritoneal folds of the primordial diaphragm become the targets of migration of muscle



Fig 3. Burden of damaging *de novo* **variants in different gene sets and sub classes of CDH.** (a) In complex cases, LGD variants were dramatically (4.6 fold) enriched in genes highly expressed (ranked in the top quartile) in mouse developing diaphragm (MDD) [<u>34</u>], and showed no enrichment in other quartiles. By comparison in isolated cases, LGD variants showed similar enrichment (~2 fold) across expression levels. (b) D-mis variants carried by complex cases also showed highest (2.4 fold) enrichment in the top quartile of MDD expression. Enrichment was evaluated by comparing observed number of variants to the baseline expectation[<u>23</u>, <u>70</u>] using a one-sided Poisson test. Bars represent the 95% confidence intervals of estimated fold enrichment.

https://doi.org/10.1371/journal.pgen.1007822.g003

progenitors, where they undergo myogenesis and morphogenesis [53]. Several related pathways were implicated including cellular response to growth factors or stress events that initiate directional migration [54], actin cytoskeletal organization and cell-cell junction assembly that drive and fine tune cell movement [55, 56]. Gene sets in protein phosphorylation and JUN--MAPK (mitogen-activated protein kinase) cascades were also enriched but not entirely due to three Noonan syndrome genes (*PTPN11*, *BRAF*, *RAF1*). The enrichment in kinase activity related pathways was supported by six novel kinase genes that overlapped with intracellular signaling functions. One kinase gene, *MAPK8IP3*, has been implicated in lung development in a mouse model [57].

Discussion

In this study, by analyzing *de novo* coding variants in CDH, we confirmed the overall enrichment of damaging *de novo* variants and identified *MYRF* as a new syndromic CDH gene. All our CDH cases with *MYRF* mutations also had CHD and most of them had genitourinary defects. The striking phenotypic similarities among the cases suggest that damaging *de novo* variants of *MYRF* disrupt the function of progenitor cells of developing diaphragm, heart and reproductive organs. In this novel *MYRF* syndrome, all cases with disease associated variants had CHD including three with Scimitar syndrome, whereas penetrance CDH was incomplete. It suggests that the manifestation of CDH in this syndrome depends on other genetic, environmental, or stochastic factors. The monozygotic twin case discordant for CDH supports that stochastic developmental events are involved.

MYRF is well known for its function in regulating myelination of the central nervous system [32]. A mouse model with conditional deletion of *MYRF* in oligodendrocyte precursors has abnormal motor skill [58]. Recently, an inherited missense variant in *MYRF* (Q403R) has



Fig 4. A functional enrichment map of genes affected by *de novo* **damaging variants in CDH.** (a) Enrichment results were visualized by a network of gene sets, where node size is proportional to the number of genes in each gene set and the thickness of edge represents the overlaps between gene sets. The significance of enrichment (p-value) is indicated by the color gradient. Functionally related gene sets are circled and manually labeled. Sub-clusters of network with similar functional annotations are grouped together as functional modules. (b) Mapping genes affected by damaging variants in CDH to the enriched functional groups shown in (a).

https://doi.org/10.1371/journal.pgen.1007822.g004

been reported as the cause of encephalopathy with reversible myelin vacuolization in a Japanese pedigree [59]. This variant is located at the same residue as the *de novo* missense variant in one of the PCGC cases but with a different substutition (Q403H). No other congenital defects were reported for the variant carriers in that family. The Q403R variant has been experimentally shown to diminish the transcription activity of a target gene [59], similar to our finding in two other missense variants (S6 Fig). Why the two different substitutions at the same amino acid position result in different phenotypes remains to be elucidated in the future. Among patients with *de novo* damaging variants in *MYRF*, one individual with the R695H variant also had intellectual disability and delayed motor skills (Table 3).

We identified 25 other individuals harboring damaging de novo variants in known or candidate DD/CHD genes, most of which have not been reported to be associated with CDH before. The significant enrichment of damaging variants among DD/CHD genes strongly suggest their causative role for majority of these cases. Similar to the case of MYRF, many DD/CHD genes have yet to be established as known disease genes. The enrichment of CDH damaging variants support their possible involvement in a broader range of developmental abnormalities which should be further evaluated in additional case cohorts with other congenital anomalies. Some recent studies of other congenital anomalies and developmental disorders have already provided further evidence for a few putative DD/CHD genes. For example, a damaging missense variant in LAMA5, a gene that plays a role in the maintenance and function of the extracellular matrix critical for pattern formation during development [60], was associated with multi-system syndrome in an Italian family [61]. Duplication of STAG2, which encodes a subunit of cohesin complex, was associated with intellectual disability and behavioral problems [62]. MEIS2 was previously nominated as a potential CDH candidate by transcriptome analysis [34] and encodes an interaction partner of transcription factor gene PBX1, haploinsufficient of which has recently been associated with multiple developmental defects including CDH [63].

Since our knowledge of DD/CHD genes is incomplete, it is possible that this observed genetic overlap represents only the tip of an iceberg. Our pathway analysis not only captured general biological process during developmental, but also identified pathways that are closely related to diaphragm development. Some novel genes prioritized by the pathway analysis have also been supported by new genetic data in other disorders. For example, *de novo* copy number loss or missense variants in *TAOK2*, one of the kinase gene implicated by the enriched gene sets of the kinase activity and MAPK signaling, has been demonstrated to cause autism and other NDD [64]. Because CDH is a relatively uncommon and lethal condition as are many other rare congenital anomalies, it is difficult to recruit large numbers of patients for genetic studies. The findings from this and other studies [15] suggest that cross-disorder analysis can be a powerful strategy for future gene discovery.

The genetic overlap between CDH and other disorders is consistent with pleiotropy among developmental disorder genes and is further supported by the highly correlated gene expression levels in multiple developing organs. We also showed that different enrichment patterns of *de novo* damaging variants between complex and isolated CDH cases is consistent with the hypothesis that variants in complex cases affect genes with more pleiotropic effects.

The pleiotropic effects of genes during development also suggest that our current classification of "isolated" cases may understate their non-diaphram abnormalities. A limitation of our study is the lack of long term clinical outcome data on many of the patients since our cohort is still relatively young. Examining the most recent medical records of patients with variants in DD/CHD genes revealed mild-to-moderate cadiovascular or NDD symptoms in several cases initially classified as isolated at birth (<u>S7 Table</u>). The medical records were often incomplete for patients who died at early infancy or were lost to follow-up (<u>Table 1</u>), and it is likely that NDD outcome in many isolated patients were underestimated [<u>65</u>, <u>66</u>]. Furthermore, almost all isolated cases also had pulmonary hypoplasia. Traditionally it was assumed that lung defects were caused by the mechanical compression by the herniated visceral, but it is clear now that development of lung and diaphargm are two intricatelly connected developmental processes [<u>67</u>], and lung defects may share common etiologies with CDH [<u>68</u>]. Among *MYRF* variant carriers, four patients who did not have diaphragm defects developed pulmonary hypoplasia (<u>Table 3</u>), further supporting common genetic control of these two processes. Larger cohorts with more detailed neurodevelopmental and long term outcomes will enhance our ability to identify additional CDH genes and provide accurate prognostic information to families to allow for future clinical diagnosis of these conditions.

In summary, our analysis of *de novo* coding variants in 362 CDH trios identified a new disease gene *MYRF*, revealed genetic overlap with other developmental disorders, and identified biological processes important for diaphragm development. Future studies will beneifit from larger sample sizes, analyzing different types of genetic variants, leveraging the information from other developmental disorders, and integrating functional genomic data.

Materials and methods

Patients recruitment

Study subjects were enrolled by the DHREAMS study (<u>http://www.cdhgenetics.com/</u>). Neonates, children and fetal cases with a diagnosis of diaphragm defects were eligible for DHREAMS. Clinical data were collected from the medical records by study personnel at each of 16 clinical sites. A complete family history of diaphragm defects and major malformations was collected on all patients by a genetic counsellor. A blood, saliva, and/or skin/diaphragm tissue sample was collected from the patient and both parents. All studies were approved by local institutional review boards, and all participants or their parents provided signed informed consent.

Cases without known pathogenic chromosome abnormalities or copy number variations [11] were selected for exome or whole-genome sequencing. A total of 283 trios with no family history of CDH with three generation and not born to consanguineous marriages were included in the current study. *De novo* coding variants on a subset trios (n = 39) have been described in our previous study [17]. In Neonates cohort, longitudinal follow-up data including Bayley III and Vineland II developmental assessments since discharge at 2 years and/or 5 years of age were gathered. Patients were evaluated to have developmental delay if at least one of the composite scores was 2 standard deviations below population average.

Patients with additional birth defects or developmental delay or other neuropsychatric phenotypes at last contact were classified as complex, and otherwise as isolated. Pulmonary hypoplasia, cardiac displacement and intestinal herniation were considered to be part of the diaphragm defect sequence and were not considered to be additional birth defects.

Subjects of BCH/MGH cohort were enrolled in "Gene Mutation and Rescue in Human Diaphragmatic Hernia" study as described previously [16]. Among 87 trios from BCH/MGH cohort, 8 trios were found to be duplicates with DHREAMS trios and were excluded from the analysis.

Whole exome/genome sequencing

Exome sequencing was performed in 79 trios that were not published before. Eleven trios were processed at the New York Genome Center. The DNA libraries were prepared using the Illumina TruSeq Sample Prep Kit (Illumina). The coding exons were captured using Agilent Sure-Select Human All Exon Kit v2 (Agilent Technologies). Samples were multiplexed and sequenced with paired-end 75bp reads on Illumina HiSeq 2500 platform according to the manufacturer's instructions. Sixty-eight trios processed at University of Washington Northwest Genome Center were captured using NimbleGen SeqCap EZ Human Exome V2 kit (Roche NimbleGen), and sequeced on HiSeq 4000 in 75 bp paired-end reads.

Another 192 trios were processed at Baylor College of Medicine Human Genome Sequencing Center using whole genome sequencing as part of the Gabriella Miller Kids First Pediatric Research Program. Among these, 27 trios were included in the previous exome study [17] but had no damaging *de novo* variants. Genomic libraries were prepared by the Illumina TruSeq DNA PCR-Free Library Prep Kit (Illumina) with average fragment length about 350 bp, and sequenced as paired-end reads of 150-bp on Illumina HiSeq X platform.

De novo variant calling and annotation

Exome and whole-genome sequencing data were processed using an inhouse pipeline implementing GATK Best Practice (version 3). Briefly, reads were mapped to human genome reference (GRCh37) using BWA-mem (version 0.7.10); duplicated reads were marked using Picard (version 1.67); variants were called using GATK (version 3.3–0) HaplotypCaller to generate gVCF files for joint genotyping. All samples within the same batch were jointly genotyped and variant quality score recalibration (VQSR) was performed using GATK. Common SNP genotypes within exome regions were used to valid parent-offspring relationships using KING (version 2.0) [69].

A variant that was presented in the offspring and had homozygous reference genotypes in both parents was considered to be a potential *de novo* variant. We used a series of stringent filters to identify *de novo* variants as described previously[70]. Briefly, we first kept variants that passed VQSR filter (tranche \leq 99.8 for SNVs and \leq 99.0 for indels) and had GATK's Fisher Strand \leq 25, quality by depth \geq 2. Then we required the candidate *de novo* variants in proband to have \geq 5 reads supporting alternative allele, \geq 20% alternative allele fraction, Phread-scaled genotype likelihood \geq 60 (GQ), and population allele frequency \leq 0.1% in ExAC; and required both parents to have > = 10 reference reads, <5% alternative allele fraction, and GQ \geq 30.

We used ANNOVAR [71] to annotate functional consequence of *de novo* variants on GEN-CODE (v19) protein coding genes. All coding *de novo* variants were manually inspected in the Integrated Genomics Viewer (http://software.broadinstitute.org/software/igv). A total of 169 variants were selected for validation using Sanger sequencing; all of them were confirmed as *de novo* variant. The number of coding *de novo* variants per proband was compared with expectations under Possion distribution.

All coding variants were classified as silent, missense, inframe, and likely-gene-disrupting (LGD, which includes frameshift indels, canonical splice site, or nonsense variants). The most severe functional effect was assigned to each variant. We defined deleterious missense variants (D-mis) by phred-scaled CADD (version 1.3) [22] score \geq 25.

De novo variant burden analysis

Baseline rate for different classes of *de novo* variants in each GENCODE coding gene were using a previously described mutation model [23, 70]. Briefly, the tri-nucleotide sequence context was used to determine the probability of each base in mutating to each other possible base

(precomputed rates are available at: <u>https://github.com/jeremymcrae/denovonear/blob/</u> <u>master/denovonear/data/rates.txt</u>). Then, the mutation rate of each functional class of point mutations in gene was calculated by adding up point mutation rates in the longest transcript. The rate of frameshift indels was presumed to be 1.1 times the nonsense mutation rate. The expected number of variants in different gene sets were calculated by summing up the classspecific variant rate in each gene in the gene set mutiplied by twice the number of patients (and if the gene is located on the non-pseudoautosomal region of chromsome X, further adjusted for female-to-male ratio [14]). The observed number of variants in each gene set and case group was then compared with the baseline expectation using Poisson test.

In burden analysis, constrained genes were defined by pLI metrics $[27] \ge 0.5$ which include a total of 5451 GENCODE genes, and all remaining genes were treated as other genes. We used a less stringent pLI threshold than previously suggested [27] for defining constrained genes, because it captured more known haploinsufficient genes important for heart and diaphragm development. Genes were also grouped by their expression levels in mouse developing diaphragm. Microarray expression profile of mouse pleuroperitoneal folds at E11.5 was taken from a previous study [34]. Normalized gene expression levels were converted to rank percentiles with smaller values corresponding to higher expression. Human orthologs of mouse genes were identified using annotations from MGI database (http://www.informatics.jax.org/). When a human gene mapped to multiple mouse genes, the highest expression level was assigned to the human gene.

RNA sequencing

Fibroblasts were obtained from diaphragm biopies at the time of diaphragm repair from 36 CDH neonatal cases most of whom carried damaging *de novo* variants, including three cases carrying *MYRF* variants (p.G81Wfs*45, V679A, and R695H). Cells were cultured in Dulbecco's Modified Eagle's Medium supplemented with 10% heat-inactivated fetal bovine serum and 1x Antibiotic/antimycotic (Gibco; Life Technologies), following standard conditions. Cells were cultured in parallel in successive passes until optimal confluence was reached, and were collected with 2.5% Trypsin (Gibco; Life Technologies) and harvested by centrifugation 5 minutes at 1200rpm. Total RNA was extracted from the cell pellet of each subject using RNeasy LipidTissue mini Kit (QIAGEN) according to manufacturer's protocol. The quality and quantity of RNA were assayed using a Qubit RNA Assay Kit in a Qubit 2.0 Fluorometer (Life Technologies) and RNA Nano 6000 Assays on a Bioanalyzer 2100 system (Agilent Technologies). cDNA libraries were prepared with the TruSeq Stranded Total RNA Sample Preparation kit (Illumina), following the manufacturer instructions. And the purified products were evaluated with an Agilent Bioanalyzer (Agilent Technologies). The library was sequenced on Illumina HiSeq 2000 platform in 100-bp paired-end reads.

RNA-seq data analysis

RNA-seq reads were mapped to the human reference genome (GRCh37) using STAR (version 2.5.2b) [72]. Gene expression levels were quantified as TPM from the output of FeatureCounts (2015–05 version) [73]. Only protein coding genes were kept for analysis and genes with no mapped reads in at least half of the samples were filter out. All sequenced samples had >20 million mapped read pairs with >90% mapping rate. Principle component (PC) analysis of gene expression profile showed that five samples were separated from others on the first two PC axes ($\underline{S5 \ Fig}$). The outlier samples were likely due to different number of passages in cell culture, and were removed from analysis.

Differential expressed genes (DEG) between cases with *MYRF* variants and others were identified using DESeq2 package [41]. DEG were selected using following criteria: adjusted p-

value < 0.5 and adjusted fold change > 0.5 or < -0.5. We noted that all three *MYRF de novo* variant carriers were male. To avoid confounding effect of gender, DEG analysis was also performed by comparing male samples with or without *MYRF* variants. The full DEG list is given in <u>S4 Data</u>.

To evaluate the consequence of *MYRF* damaging variants on patients' transcriptome, we tested if putative MYRF target genes were systematically down-regulated in the fibroblast cells with *MYRF* variants using gene set enrichment analysis (GSEA). The MYRF target genes as oligodendrocyte-specific genes that had at least one MYRF ChIP-seq binding peaks with 100kb of transcription start site [40]. We then identified corresponding human orthologs using biomaRt package [74]. A total of 74 human genes were defined as putative target genes for GSEA.

Quantitative PCR

We selected six genes from differentially expressed genes between *MYRF* mutation carriers and other cases, including four down-regulated (*GATA4*, *DBNDD2*, *MYO1D* and *NFASC*) and two up-regulated (*H3F3C* and *SEMA3A*) in *MYRF* mutant cells. First-strand cDNA was synthesized from the total RNA (500ng~1 µg) using the RNA to cDNA EcoDry Premix (Random Hexamers) kit (TaKaRa) according to manufacturer's instructions. Primers for the selected genes (<u>S6 Table</u>) were synthesized by IdtDNA. All qPCR reactions were performed in a total of 10 µl volume, comprising 5 µl 2x SYBR Green I Master Mix (Promega), 1 µl 10nM of each primer and 2 µl of 1:20 diluted cDNA in 96-well plates using CFX Connect Real-Time PCR Detection System (Bio-Rad). All reactions were performed in triplicate and the conditions were 5 minutes at 95°C, then 40 cycles of 95°C at 15 seconds and 60°C at 30 seconds. The relative expression levels were calculated using the standard curve method relative to the β -actin housekeeping gene. Five-serial 4-fold dilutions of cDNA samples were used to construct the standard curves for each primer.

Cross-disorder genetic overlap

To assess the genetic overlap with other developmental disorders and especially CHD, we tested if the *de novo* damaging variants in CDH cases were enriched in known and putative CHD and DD genes. DD genes were extracted from DDG2P database [75] (accessed on Jan 11, 2018) and filtered to keep "allelic requirement" as monoallelic, X-linked dominant or hemizygous, and required "organ specificity list" to include brain, heart or not specific to any organ. A total 508 DD genes were identified, including 460 confirmed DD genes. CHD genes were collected based on a recent exome study of 2645 trios [29]. CHD genes included high heart expressed genes (HHE; ranked at top 25%) or known human CHD genes that were affected by more than one damaging de novo variants (LGD or D-mis defined by meta-SVM [76] as the original publication on CHD [29]) or constrained (pLI \geq 0.5) HHE genes affected by only one damaing variants from the same study. A total 200 CHD genes were identified, 57 of which overlapped with DD genes.

To assess if the exome-wide *de novo* damaging variants in CDH were enriched in CHD and DD genes, simulations were done to randomly place variants to the coding regions in a way that keeps the number of variants, tri-nucleotide context, functional effect, and deleteriouness prediction the same as that of the observed data [77]. Here the coding region was defined as coding sequences and canonical splice sites of all GENCODE v19 coding genes. For damaging mutations identified from WES data, the coding regions were restricted to the regions that have at >10X coverage in least 80% samples. Empirical p-value was calculated as the chance when there were more simulated damaging variants than observed in the given gene set. We

ran 50,000 simulations to evaluate the significance. And the expected number of variants in a gene set was the average number of randomly generated variants in a gene set over all simulations.

Functional enrichment map

To evaluate the functional convergence of genes affected by damaging variants, we extracted 89 genes that included 86 constrained genes (pLI \geq 0.5), two known candidates for CDH (*GATA6*, *WT1*), and a known haploinsufficient gene (*KDM5B*). Gene sets were derived from Gene Ontology Biological Process (GO-BP, accessed Feb 1st, 2018). The GO-BO categories that were statistically over-represented in the gene list (FDR<0.1) were identified using hyper-geometric test implemented by BINGO [78]. Terms annotating more than 750 or less than 25 genes were discarded, because large gene-sets usually represent broad categories without specific biological meaning. Small gene sets on the other hand are not likely to produce statistically significant results.

Enriched gene sets were graphically visualized as a network, in which each gene set is a node and edges represent overlap between sets. The Cytoscape software [79] and Enrichment-Map plugin [80] were used to construct the network. The color gradient of nodes reflects the enrichment p-values. Node size is proportional to the number of genes in the gene set. Edge thickness is proportional to the similarity score between gene sets which is defined by the average of Jaccard coefficient and overlap coefficient [80]. Enriched gene sets with highly overlapping genes (S6 Data) were grouped together and annotated manually.

Supporting information

S1 Fig. Depth of coverage. Scatter plot and marginal histograms for mean depth and percentages of targeted regions with at least 10 or 15 reads are shown for whole-exome sequencing (a) and whole-genome sequening samples (b). (PDF)

S2 Fig. Distribution of *de novo* **coding variants per proband.** Distribution of number of de novo coding variants per proband in whole-exome (a and b) and whole-genome sequenced trios (c).

(PDF)

S2 Fig. Multiple sequence alignment of the DBD domain (a) and ICA domain (b) of the MYRF protein.

(PDF)

S4 Fig. The predicted effects of *de novo* **missense variants MYRF 3D structure.** The predicated local 3D structures of wild and mutant type proteins are shown for F387S (a), Q403H (b), G435R (c), L479R (d), and R695H (e). V679A cannot be modeled due to lack of homologues template. (PDF)

S5 Fig. Principle component analysis of RNA-seq samples before (a) and after (b) removing outliers.

(PDF)

S6 Fig. (Related to Fig1b) The impact of *de novo* missense variants on the patient transcriptomes. (a) The distribution of mean z-scores of gene expression. (b) Gene set enrichment analysis of MYRF target genes. (PDF) **S7 Fig. qPCR validation of selected genes differentially expressed between MYRF mutation carriers and other cases.** The relative expression levels of six selected genes from qPCR (a) are compared with the TPM metrics of RNAseq (b). (PDF)

S8 Fig. Expression trajectories of *MYRF* and *GATA4* in mouse developing diaphragm and lung.

(PDF)

S9 Fig. Enrichment of damaging variants in DD/CHD genes that have not been implicated in CDH. (a) Comparing the observed vs expected number of damaging variants on DD/CHD genes after excluding known CDH candidate genes. (b) The same as (a) but using all constrained genes as the background. (PDF)

S1 Data. Demographic and clinical characteristics of cases. (XLSX)

S2 Data. Full list of de novo coding variants. (XLSX)

S3 Data. Putative MYRF target genes. (XLSX)

S4 Data. Differentially expressed genes between *MYRF* variant carriers and other cases. (XLSX)

S5 Data. Gene sets used in cross-disorder analysis. (XLSX)

S6 Data. (Related to Fig4) Enriched Gene Ontology terms in the functional map. (XLSX)

S1 Table. Sequencing summary. (PDF)

S2 Table. (Related to Table 2) Burden of *de novo* variants in different sub-groups of patients.

(PDF)

S3 Table. Genes affected by multiple *de novo* functional variants. (PDF)

S4 Table. Pathogenicity prediction of *MYRF de novo* missense variants. (PDF)

S5 Table. LGD variants of *MYRF* in general populations. (PDF)

S6 Table. Primers for qPCR validation of selected differentially expressed genes. (PDF)

S7 Table. (Related to Fig 2C) Clinical information of cases carrying damaging variants in DD/CHD genes. (PDF) **S1 Text. Supplementary references.** (PDF)

Acknowledgments

We would like to thank the patients and their families for their generous contribution. We are grateful for the technical assistance provided by Patricia Lanzano, Jiancheng Guo, Suying Bao, and Liyong Deng from Columbia University, Jessica Kim at Boston Children's Hospital, and Caroline Coletti and Pooja Bhayani at Massachusetts General Hospital. We thank our clinical coordinators across the DHREAMS centers: Trish Burns at Cincinnati Children's Hospital, Sheila Horak at Children's Hospital & Medical Center of Omaha, Brandy Gonzales at Oregon Health and Science University, Karen Lukas at St. Louis Children's Hospital, Jeannie Kreutzman at CS Mott Children's Hospital, Min Shi at Children's Hospital of Pittsburgh, Michelle Knezevich and Cheryl Kornberg at Medical College of Wisconsin.

Author Contributions

Conceptualization: Wendy K. Chung, Yufeng Shen.

Data curation: Lan Yu, Julia Wynn, Rebecca Hernan.

Formal analysis: Hongjian Qi, Xueya Zhou, Haoquan Zhao, Yicheng Guo, Na Zhu, Alexander Kitaygorodsky, Yufeng Shen.

Funding acquisition: Patricia K. Donahoe, Wendy K. Chung, Yufeng Shen.

Methodology: Hongjian Qi, Lan Yu, Xueya Zhou, Yufeng Shen.

Project administration: Wendy K. Chung, Yufeng Shen.

Resources: Gudrun Aspelund, Foong-Yen Lim, Timothy Crombleholme, Robert Cusick, Kenneth Azarow, Melissa E. Danko, Dai Chung, Brad W. Warner, George B. Mychaliska, Douglas Potoka, Amy J. Wagner, Mahmoud ElFiky, Jay M. Wilson, Debbie Nickerson, Michael Bamshad, Frances A. High, Mauro Longoni, Patricia K. Donahoe.

Supervision: Wendy K. Chung, Yufeng Shen.

Validation: Lan Yu.

Visualization: Xueya Zhou, Haoquan Zhao, Yicheng Guo.

Writing - original draft: Hongjian Qi, Lan Yu, Xueya Zhou.

Writing - review & editing: Wendy K. Chung, Yufeng Shen.

References

- Chandrasekharan PK, Rawat M, Madappa R, Rothstein DH, Lakshminrusimha S. Congenital Diaphragmatic hernia—a review. Matern Health Neonatol Perinatol. 2017; 3:6. <u>https://doi.org/10.1186/s40748-017-0045-1</u> PMID: <u>28331629</u>; PubMed Central PMCID: PMCPMC5356475.
- Pober BR, Russell MK, Ackerman KG. Congenital Diaphragmatic Hernia Overview. In: Pagon RA, Adam MP, Ardinger HH, Wallace SE, Amemiya A, Bean LJH, et al., editors. GeneReviews(R). Seattle (WA)2010.
- Zaiss I, Kehl S, Link K, Neff W, Schaible T, Sutterlin M, et al. Associated malformations in congenital diaphragmatic hernia. Am J Perinatol. 2011; 28(3):211–8. Epub 2010/10/28. <u>https://doi.org/10.1055/s-0030-1268235</u> PMID: 20979012.
- Stoll C, Alembik Y, Dott B, Roth MP. Associated malformations in cases with congenital diaphragmatic hernia. Genet Couns. 2008; 19(3):331–9. Epub 2008/11/11. PMID: <u>18990989</u>.

- Zalla JM, Stoddard GJ, Yoder BA. Improved mortality rate for congenital diaphragmatic hernia in the modern era of management: 15 year experience in a single institution. J Pediatr Surg. 2015; 50(4):524– 7. <u>https://doi.org/10.1016/j.jpedsurg.2014.11.002</u> PMID: <u>25840055</u>; PubMed Central PMCID: PMCPMC4386064.
- Coughlin MA, Werner NL, Gajarski R, Gadepalli S, Hirschl R, Barks J, et al. Prenatally diagnosed severe CDH: mortality and morbidity remain high. J Pediatr Surg. 2016; 51(7):1091–5. Epub 2015/12/ 15. <u>https://doi.org/10.1016/j.jpedsurg.2015.10.082</u> PMID: <u>26655216</u>.
- Pober BR, Lin A, Russell M, Ackerman KG, Chakravorty S, Strauss B, et al. Infants with Bochdalek diaphragmatic hernia: sibling precurrence and monozygotic twin discordance in a hospital-based malformation surveillance program. Am J Med Genet A. 2005; 138A(2):81–8. Epub 2005/08/12. <u>https://doi.org/10.1002/ajmg.a.30904</u> PMID: <u>16094667</u>; PubMed Central PMCID: PMCPMC2891716.
- Slavotinek AM. Single gene disorders associated with congenital diaphragmatic hernia. Am J Med Genet C Semin Med Genet. 2007; 145C(2):172–83. <u>https://doi.org/10.1002/ajmg.c.30125</u> PMID: <u>17436300</u>.
- Pober BR. Overview of epidemiology, genetics, birth defects, and chromosome abnormalities associated with CDH. Am J Med Genet C Semin Med Genet. 2007; 145C(2):158–71. https://doi.org/10.1002/ajmg.c.30126 PMID: <u>17436298</u>; PubMed Central PMCID: PMCPMC2891729.
- Brady PD, DeKoninck P, Fryns JP, Devriendt K, Deprest JA, Vermeesch JR. Identification of dosagesensitive genes in fetuses referred with severe isolated congenital diaphragmatic hernia. Prenat Diagn. 2013; 33(13):1283–92. Epub 2013/10/15. <u>https://doi.org/10.1002/pd.4244</u> PMID: <u>24122781</u>.
- Yu L, Wynn J, Ma L, Guha S, Mychaliska GB, Crombleholme TM, et al. De novo copy number variants are associated with congenital diaphragmatic hernia. J Med Genet. 2012; 49(10):650–9. <u>https://doi.org/ 10.1136/jmedgenet-2012-101135</u> PMID: 23054247; PubMed Central PMCID: PMCPMC3696999.
- Zhu Q, High FA, Zhang C, Cerveira E, Russell MK, Longoni M, et al. Systematic analysis of copy number variation associated with congenital diaphragmatic hernia. Proc Natl Acad Sci U S A. 2018; 115 (20):5247–52. Epub 2018/05/02. <u>https://doi.org/10.1073/pnas.1714885115</u> PMID: <u>29712845</u>; PubMed Central PMCID: PMCPMC5960281.
- **13.** Beurskens N, Klaassens M, Rottier R, de Klein A, Tibboel D. Linking animal models to human congenital diaphragmatic hernia. Birth Defects Res A Clin Mol Teratol. 2007; 79(8):565–72. Epub 2007/05/01. <u>https://doi.org/10.1002/bdra.20370</u> PMID: <u>17469205</u>.
- Deciphering Developmental Disorders S. Large-scale discovery of novel genetic causes of developmental disorders. Nature. 2015; 519(7542):223–8. Epub 2014/12/24. <u>https://doi.org/10.1038/nature14135 PMID: 25533962</u>.
- Deciphering Developmental Disorders S. Prevalence and architecture of de novo mutations in developmental disorders. Nature. 2017; 542(7642):433–8. Epub 2017/01/31. <u>https://doi.org/10.1038/nature21062 PMID: 28135719</u>.
- Longoni M, High FA, Qi H, Joy MP, Hila R, Coletti CM, et al. Genome-wide enrichment of damaging de novo variants in patients with isolated and complex congenital diaphragmatic hernia. Hum Genet. 2017; 136(6):679–91. <u>https://doi.org/10.1007/s00439-017-1774-y</u> PMID: <u>28303347</u>; PubMed Central PMCID: PMCPMC5453716.
- Yu L, Sawle AD, Wynn J, Aspelund G, Stolar CJ, Arkovitz MS, et al. Increased burden of de novo predicted deleterious variants in complex congenital diaphragmatic hernia. Hum Mol Genet. 2015; 24 (16):4764–73. <u>https://doi.org/10.1093/hmg/ddv196</u> PMID: <u>26034137</u>; PubMed Central PMCID: PMCPMC4512631.
- Meynert AM, Ansari M, FitzPatrick DR, Taylor MS. Variant detection sensitivity and biases in whole genome and exome sequencing. BMC Bioinformatics. 2014; 15:247. <u>https://doi.org/10.1186/1471-</u> 2105-15-247 PMID: 25038816; PubMed Central PMCID: PMCPMC4122774.
- Belkadi A, Bolze A, Itan Y, Cobat A, Vincent QB, Antipenko A, et al. Whole-genome sequencing is more powerful than whole-exome sequencing for detecting exome variants. Proc Natl Acad Sci U S A. 2015; 112(17):5473–8. Epub 2015/04/02. <u>https://doi.org/10.1073/pnas.1418631112</u> PMID: <u>25827230</u>; PubMed Central PMCID: PMCPMC4418901.
- Hinton CF, Siffel C, Correa A, Shapira SK. Survival Disparities Associated with Congenital Diaphragmatic Hernia. Birth Defects Res. 2017; 109(11):816–23. Epub 2017/04/12. <u>https://doi.org/10.1002/</u>bdr2.1015 PMID: 28398654; PubMed Central PMCID: PMCPMC5507197.
- Leeuwen L, Mous DS, van Rosmalen J, Olieman JF, Andriessen L, Gischler SJ, et al. Congenital Diaphragmatic Hernia and Growth to 12 Years. Pediatrics. 2017;140(2). Epub 2017/07/16. <u>https://doi.org/ 10.1542/peds.2016-3659</u> PMID: <u>28710247</u>.
- 22. Kircher M, Witten DM, Jain P, O'Roak BJ, Cooper GM, Shendure J. A general framework for estimating the relative pathogenicity of human genetic variants. Nat Genet. 2014; 46(3):310–5. Epub 2014/02/04. <u>https://doi.org/10.1038/ng.2892</u> PMID: 24487276; PubMed Central PMCID: PMCPMC3992975.

- 23. Samocha KE, Robinson EB, Sanders SJ, Stevens C, Sabo A, McGrath LM, et al. A framework for the interpretation of de novo mutation in human disease. Nat Genet. 2014; 46(9):944–50. <u>https://doi.org/10.1038/ng.3050</u> PMID: <u>25086666</u>; PubMed Central PMCID: PMCPMC4222185.
- Jacquemont S, Coe BP, Hersch M, Duyzend MH, Krumm N, Bergmann S, et al. A higher mutational burden in females supports a "female protective model" in neurodevelopmental disorders. Am J Hum Genet. 2014; 94(3):415–25. Epub 2014/03/04. <u>https://doi.org/10.1016/j.ajhg.2014.02.001</u> PMID: <u>24581740</u>; PubMed Central PMCID: PMCPMC3951938.
- 25. Wang B, Ji T, Zhou X, Wang J, Wang X, Wang J, et al. CNV analysis in Chinese children of mental retardation highlights a sex differentiation in parental contribution to de novo and inherited mutational burdens. Sci Rep. 2016; 6:25954. Epub 2016/06/04. <u>https://doi.org/10.1038/srep25954</u> PMID: <u>27257017</u>; PubMed Central PMCID: PMCPMC4891738.
- Kosmicki JA, Samocha KE, Howrigan DP, Sanders SJ, Slowikowski K, Lek M, et al. Refining the role of de novo protein-truncating variants in neurodevelopmental disorders by using population reference samples. Nat Genet. 2017; 49(4):504–10. Epub 2017/02/14. <u>https://doi.org/10.1038/ng.3789</u> PMID: 28191890; PubMed Central PMCID: PMCPMC5496244.
- Lek M, Karczewski KJ, Minikel EV, Samocha KE, Banks E, Fennell T, et al. Analysis of protein-coding genetic variation in 60,706 humans. Nature. 2016; 536(7616):285–91. <u>https://doi.org/10.1038/ nature19057</u> PMID: 27535533; PubMed Central PMCID: PMCPMC5018207.
- Ware JS, Samocha KE, Homsy J, Daly MJ. Interpreting de novo Variation in Human Disease Using denovolyzeR. Curr Protoc Hum Genet. 2015; 87:7 25 1–15. <u>https://doi.org/10.1002/0471142905.</u> hg0725s87 PMID: 26439716; PubMed Central PMCID: PMCPMC4606471.
- Jin SC, Homsy J, Zaidi S, Lu Q, Morton S, DePalma SR, et al. Contribution of rare inherited and de novo variants in 2,871 congenital heart disease probands. Nat Genet. 2017; 49(11):1593–601. Epub 2017/10/11. <u>https://doi.org/10.1038/ng.3970</u> PMID: <u>28991257</u>; PubMed Central PMCID: PMCPMC5675000.
- Pinz H, Pyle LC, Li D, Izumi K, Skraban C, Tarpinian J, et al. De novo variants in Myelin regulatory factor (MYRF) as candidates of a new syndrome of cardiac and urogenital anomalies. Am J Med Genet A. 2018; 176(4):969–72. Epub 2018/02/16. <u>https://doi.org/10.1002/ajmg.a.38620</u> PMID: <u>29446546</u>; PubMed Central PMCID: PMCPMC5867271.
- Chitayat D, Shannon P, Uster T, Nezarati MM, Schnur RE, Bhoj EJ. An Additional Individual with a De Novo Variant in Myelin Regulatory Factor (MYRF) with Cardiac and Urogenital Anomalies: Further Proof of Causality: Comments on the article by Pinz et al. (). Am J Med Genet A. 2018. Epub 2018/08/ 03. https://doi.org/10.1002/ajmg.a.40360 PMID: 30070761.
- Emery B, Agalliu D, Cahoy JD, Watkins TA, Dugas JC, Mulinyawe SB, et al. Myelin gene regulatory factor is a critical transcriptional regulator required for CNS myelination. Cell. 2009; 138(1):172–85. Epub 2009/07/15. <u>https://doi.org/10.1016/j.cell.2009.04.031</u> PMID: <u>19596243</u>; PubMed Central PMCID: PMCPMC2757090.
- Hornig J, Frob F, Vogl MR, Hermans-Borgmeyer I, Tamm ER, Wegner M. The transcription factors Sox10 and Myrf define an essential regulatory network module in differentiating oligodendrocytes. PLoS Genet. 2013; 9(10):e1003907. Epub 2013/11/10. <u>https://doi.org/10.1371/journal.pgen.1003907</u> PMID: <u>24204311</u>; PubMed Central PMCID: PMCPMC3814293.
- Russell MK, Longoni M, Wells J, Maalouf FI, Tracy AA, Loscertales M, et al. Congenital diaphragmatic hernia candidate genes derived from embryonic transcriptomes. Proc Natl Acad Sci U S A. 2012; 109 (8):2978–83. <u>https://doi.org/10.1073/pnas.1121621109</u> PMID: <u>22315423</u>; PubMed Central PMCID: PMCPMC3286948.
- Zaidi S, Choi M, Wakimoto H, Ma L, Jiang J, Overton JD, et al. De novo mutations in histone-modifying genes in congenital heart disease. Nature. 2013; 498(7453):220–3. Epub 2013/05/15. <u>https://doi.org/ 10.1038/nature12141</u> PMID: <u>23665959</u>; PubMed Central PMCID: PMCPMC3706629.
- Li Z, Park Y, Marcotte EM. A Bacteriophage tailspike domain promotes self-cleavage of a human membrane-bound transcription factor, the myelin regulatory factor MYRF. PLoS Biol. 2013; 11(8):e1001624. Epub 2013/08/24. <u>https://doi.org/10.1371/journal.pbio.1001624</u> PMID: <u>23966832</u>; PubMed Central PMCID: PMCPMC3742443.
- Samocha KE, Kosmicki JA, Karczewski KJ, O'Donnell-Luria AH, Pierce-Hoffman E, MacArthur DG, et al. Regional missense constraint improves variant deleteriousness prediction. bioRxiv. 2017. <u>https:// doi.org/10.1101/148353</u>
- Kim D, Choi JO, Fan C, Shearer RS, Sharif M, Busch P, et al. Homo-trimerization is essential for the transcription factor function of Myrf for oligodendrocyte differentiation. Nucleic Acids Res. 2017; 45 (9):5112–25. Epub 2017/02/06. <u>https://doi.org/10.1093/nar/gkx080</u> PMID: <u>28160598</u>; PubMed Central PMCID: PMCPMC5436001.

- Schulz EC, Dickmanns A, Urlaub H, Schmitt A, Muhlenhoff M, Stummeyer K, et al. Crystal structure of an intramolecular chaperone mediating triple-beta-helix folding. Nat Struct Mol Biol. 2010; 17(2):210–5. Epub 2010/02/02. <u>https://doi.org/10.1038/nsmb.1746</u> PMID: 20118935.
- Bujalka H, Koenning M, Jackson S, Perreau VM, Pope B, Hay CM, et al. MYRF is a membrane-associated transcription factor that autoproteolytically cleaves to directly activate myelin genes. PLoS Biol. 2013; 11(8):e1001625. Epub 2013/08/24. <u>https://doi.org/10.1371/journal.pbio.1001625</u> PMID: 23966833; PubMed Central PMCID: PMCPMC3742440.
- Love MI, Huber W, Anders S. Moderated estimation of fold change and dispersion for RNA-seq data with DESeq2. Genome Biol. 2014; 15(12):550. Epub 2014/12/18. <u>https://doi.org/10.1186/s13059-014-0550-8</u> PMID: 25516281; PubMed Central PMCID: PMCPMC4302049.
- 42. Subramanian A, Tamayo P, Mootha VK, Mukherjee S, Ebert BL, Gillette MA, et al. Gene set enrichment analysis: a knowledge-based approach for interpreting genome-wide expression profiles. Proc Natl Acad Sci U S A. 2005; 102(43):15545–50. Epub 2005/10/04. <u>https://doi.org/10.1073/pnas.0506580102</u> PMID: <u>16199517</u>; PubMed Central PMCID: PMCPMC1239896.
- 43. Yu L, Wynn J, Cheung YH, Shen Y, Mychaliska GB, Crombleholme TM, et al. Variants in GATA4 are a rare cause of familial and sporadic congenital diaphragmatic hernia. Hum Genet. 2013; 132(3):285–92. Epub 2012/11/10. <u>https://doi.org/10.1007/s00439-012-1249-0</u> PMID: <u>23138528</u>; PubMed Central PMCID: PMCPMC3570587.
- Iossifov I, O'Roak BJ, Sanders SJ, Ronemus M, Krumm N, Levy D, et al. The contribution of de novo coding mutations to autism spectrum disorder. Nature. 2014; 515(7526):216–21. Epub 2014/11/05. <u>https://doi.org/10.1038/nature13908</u> PMID: 25363768; PubMed Central PMCID: PMCPMC4313871.
- Ackerman KG, Herron BJ, Vargas SO, Huang H, Tevosian SG, Kochilas L, et al. Fog2 is required for normal diaphragm and lung development in mice and humans. PLoS Genet. 2005; 1(1):58–65. Epub 2005/08/17. <u>https://doi.org/10.1371/journal.pgen.0010010</u> PMID: <u>16103912</u>; PubMed Central PMCID: PMCPMC1183529.
- Yu L, Bennett JT, Wynn J, Carvill GL, Cheung YH, Shen Y, et al. Whole exome sequencing identifies de novo mutations in GATA6 associated with congenital diaphragmatic hernia. J Med Genet. 2014; 51 (3):197–202. Epub 2014/01/05. <u>https://doi.org/10.1136/jmedgenet-2013-101989</u> PMID: <u>24385578</u>; PubMed Central PMCID: PMCPMC3955383.
- Longoni M, High FA, Russell MK, Kashani A, Tracy AA, Coletti CM, et al. Molecular pathogenesis of congenital diaphragmatic hernia revealed by exome sequencing, developmental data, and bioinformatics. Proc Natl Acad Sci U S A. 2014; 111(34):12450–5. Epub 2014/08/12. <u>https://doi.org/10.1073/pnas.</u> <u>1412509111</u> PMID: <u>25107291</u>; PubMed Central PMCID: PMCPMC4151769.
- Richards S, Aziz N, Bale S, Bick D, Das S, Gastier-Foster J, et al. Standards and guidelines for the interpretation of sequence variants: a joint consensus recommendation of the American College of Medical Genetics and Genomics and the Association for Molecular Pathology. Genet Med. 2015; 17(5):405–24. Epub 2015/03/06. <u>https://doi.org/10.1038/gim.2015.30</u> PMID: <u>25741868</u>; PubMed Central PMCID: PMCPMC4544753.
- Bielinska M, Jay PY, Erlich JM, Mannisto S, Urban Z, Heikinheimo M, et al. Molecular genetics of congenital diaphragmatic defects. Ann Med. 2007; 39(4):261–74. Epub 2007/06/15. <u>https://doi.org/10.</u> 1080/07853890701326883 PMID: <u>17558598</u>; PubMed Central PMCID: PMCPMC2174621.
- Jay PY, Bielinska M, Erlich JM, Mannisto S, Pu WT, Heikinheimo M, et al. Impaired mesenchymal cell function in Gata4 mutant mice leads to diaphragmatic hernias and primary lung defects. Dev Biol. 2007; 301(2):602–14. Epub 2006/10/31. <u>https://doi.org/10.1016/j.ydbio.2006.09.050</u> PMID: <u>17069789</u>; PubMed Central PMCID: PMCPMC1808541.
- Seidman JG, Seidman C. Transcription factor haploinsufficiency: when half a loaf is not enough. J Clin Invest. 2002; 109(4):451–5. Epub 2002/02/21. <u>https://doi.org/10.1172/JCl15043</u> PMID: <u>11854316</u>; PubMed Central PMCID: PMCPMC150881.
- Bjornsson HT. The Mendelian disorders of the epigenetic machinery. Genome Res. 2015; 25 (10):1473–81. Epub 2015/10/03. <u>https://doi.org/10.1101/gr.190629.115</u> PMID: <u>26430157</u>; PubMed Central PMCID: PMCPMC4579332.
- Merrell AJ, Kardon G. Development of the diaphragm—a skeletal muscle essential for mammalian respiration. FEBS J. 2013; 280(17):4026–35. Epub 2013/04/17. <u>https://doi.org/10.1111/febs.12274</u> PMID: 23586979; PubMed Central PMCID: PMCPMC3879042.
- Parsons JT, Horwitz AR, Schwartz MA. Cell adhesion: integrating cytoskeletal dynamics and cellular tension. Nat Rev Mol Cell Biol. 2010; 11(9):633–43. Epub 2010/08/24. <u>https://doi.org/10.1038/nrm2957</u> PMID: 20729930; PubMed Central PMCID: PMCPMC2992881.
- Pollard TD, Borisy GG. Cellular motility driven by assembly and disassembly of actin filaments. Cell. 2003; 112(4):453–65. Epub 2003/02/26. PMID: <u>12600310</u>.

- 56. Friedl P, Mayor R. Tuning Collective Cell Migration by Cell-Cell Junction Regulation. Cold Spring Harb Perspect Biol. 2017; 9(4). Epub 2017/01/18. <u>https://doi.org/10.1101/cshperspect.a029199</u> PMID: 28096261.
- Ha HY, Kim JB, Cho IH, Joo HJ, Kim KS, Lee KW, et al. Morphogenetic lung defects of JSAP1-deficient embryos proceeds via the disruptions of the normal expressions of cytoskeletal and chaperone proteins. Proteomics. 2008; 8(5):1071–80. Epub 2008/03/08. <u>https://doi.org/10.1002/pmic.200700815</u> PMID: 18324732.
- Xiao L, Ohayon D, McKenzie IA, Sinclair-Wilson A, Wright JL, Fudge AD, et al. Rapid production of new oligodendrocytes is required in the earliest stages of motor-skill learning. Nat Neurosci. 2016; 19 (9):1210–7. Epub 2016/07/28. <u>https://doi.org/10.1038/nn.4351</u> PMID: <u>27455109</u>; PubMed Central PMCID: PMCPMC5008443.
- Kurahashi H, Azuma Y, Masuda A, Okuno T, Nakahara E, Imamura T, et al. MYRF is associated with encephalopathy with reversible myelin vacuolization. Ann Neurol. 2018; 83(1):98–106. Epub 2017/12/ 22. https://doi.org/10.1002/ana.25125 PMID: 29265453.
- Spenle C, Simon-Assmann P, Orend G, Miner JH. Laminin alpha5 guides tissue patterning and organogenesis. Cell Adh Migr. 2013; 7(1):90–100. Epub 2012/10/19. <u>https://doi.org/10.4161/cam.22236</u> PMID: 23076210; PubMed Central PMCID: PMCPMC3544791.
- Sampaolo S, Napolitano F, Tirozzi A, Reccia MG, Lombardi L, Farina O, et al. Identification of the first dominant mutation of LAMA5 gene causing a complex multisystem syndrome due to dysfunction of the extracellular matrix. J Med Genet. 2017; 54(10):710–20. Epub 2017/07/25. <u>https://doi.org/10.1136/ jmedgenet-2017-104555</u> PMID: 28735299.
- Kumar R, Corbett MA, Van Bon BW, Gardner A, Woenig JA, Jolly LA, et al. Increased STAG2 dosage defines a novel cohesinopathy with intellectual disability and behavioral problems. Hum Mol Genet. 2015; 24(25):7171–81. Epub 2015/10/08. <u>https://doi.org/10.1093/hmg/ddv414</u> PMID: <u>26443594</u>.
- 63. Slavotinek A, Risolino M, Losa M, Cho MT, Monaghan KG, Schneidman-Duhovny D, et al. De novo, deleterious sequence variants that alter the transcriptional activity of the homeoprotein PBX1 are associated with intellectual disability and pleiotropic developmental defects. Hum Mol Genet. 2017; 26 (24):4849–60. Epub 2017/10/17. https://doi.org/10.1093/hmg/ddx363 PMID: 29036646.
- Richter M, Murtaza N, Scharrenberg R, White SH, Johanns O, Walker S, et al. Altered TAOK2 activity causes autism-related neurodevelopmental and cognitive abnormalities through RhoA signaling. Mol Psychiatry. 2018. Epub 2018/02/23. <u>https://doi.org/10.1038/s41380-018-0025-5</u> PMID: <u>29467497</u>.
- 65. Danzer E, Gerdes M, Bernbaum J, D'Agostino J, Bebbington MW, Siegle J, et al. Neurodevelopmental outcome of infants with congenital diaphragmatic hernia prospectively enrolled in an interdisciplinary follow-up program. J Pediatr Surg. 2010; 45(9):1759–66. <u>https://doi.org/10.1016/j.jpedsurg.2010.03.011</u> PMID: 20850617
- 66. Danzer E, Hoffman C, D'Agostino JA, Gerdes M, Bernbaum J, Antiel RM, et al. Neurodevelopmental outcomes at 5 years of age in congenital diaphragmatic hernia. J Pediatr Surg. 2017; 52(3):437–43. Epub 2016/09/14. <u>https://doi.org/10.1016/i.jpedsurg.2016.08.008</u> PMID: 27622588
- Shannon JM, Hyatt BA. Epithelial-mesenchymal interactions in the developing lung. Annu Rev Physiol. 2004; 66:625–45. Epub 2004/02/24. <u>https://doi.org/10.1146/annurev.physiol.66.032102.135749</u> PMID: <u>14977416</u>
- Keijzer R, Liu J, Deimling J, Tibboel D, Post M. Dual-hit hypothesis explains pulmonary hypoplasia in the nitrofen model of congenital diaphragmatic hernia. Am J Pathol. 2000; 156(4):1299–306. Epub 2000/04/07. <u>https://doi.org/10.1016/S0002-9440(10)65000-6</u> PubMed Central PMCID: PMCPMC1876880. PMID: <u>10751355</u>
- Manichaikul A, Mychaleckyj JC, Rich SS, Daly K, Sale M, Chen WM. Robust relationship inference in genome-wide association studies. Bioinformatics. 2010; 26(22):2867–73. Epub 2010/10/12. <u>https://doi.org/10.1093/bioinformatics/btg559</u> PMID: 20926424; PubMed Central PMCID: PMCPMC3025716.
- Homsy J, Zaidi S, Shen Y, Ware JS, Samocha KE, Karczewski KJ, et al. De novo mutations in congenital heart disease with neurodevelopmental and other congenital anomalies. Science. 2015; 350 (6265):1262–6. <u>https://doi.org/10.1126/science.aac9396</u> PMID: <u>26785492</u>; PubMed Central PMCID: PMCPMC4890146.
- Wang K, Li M, Hakonarson H. ANNOVAR: functional annotation of genetic variants from high-throughput sequencing data. Nucleic Acids Res. 2010; 38(16):e164. Epub 2010/07/06. <u>https://doi.org/10.1093/</u> nar/gkq603 PMID: 20601685; PubMed Central PMCID: PMCPMC2938201.
- 72. Dobin A, Davis CA, Schlesinger F, Drenkow J, Zaleski C, Jha S, et al. STAR: ultrafast universal RNAseq aligner. Bioinformatics. 2013; 29(1):15–21. Epub 2012/10/30. <u>https://doi.org/10.1093/</u> <u>bioinformatics/bts635</u> PMID: 23104886; PubMed Central PMCID: PMCPMC3530905.

- **73.** Liao Y, Smyth GK, Shi W. featureCounts: an efficient general purpose program for assigning sequence reads to genomic features. Bioinformatics. 2014; 30(7):923–30. Epub 2013/11/15. <u>https://doi.org/10.1093/bioinformatics/btt656</u> PMID: 24227677.
- 74. Durinck S, Moreau Y, Kasprzyk A, Davis S, De Moor B, Brazma A, et al. BioMart and Bioconductor: a powerful link between biological databases and microarray data analysis. Bioinformatics. 2005; 21 (16):3439–40. Epub 2005/08/06. <u>https://doi.org/10.1093/bioinformatics/bti525</u> PMID: <u>16082012</u>.
- 75. Wright CF, Fitzgerald TW, Jones WD, Clayton S, McRae JF, van Kogelenberg M, et al. Genetic diagnosis of developmental disorders in the DDD study: a scalable analysis of genome-wide research data. Lancet. 2015; 385(9975):1305–14. Epub 2014/12/23. <u>https://doi.org/10.1016/S0140-6736(14)61705-0</u> PMID: <u>25529582</u>; PubMed Central PMCID: PMCPMC4392068.
- 76. Dong C, Wei P, Jian X, Gibbs R, Boerwinkle E, Wang K, et al. Comparison and integration of deleteriousness prediction methods for nonsynonymous SNVs in whole exome sequencing studies. Hum Mol Genet. 2015; 24(8):2125–37. Epub 2015/01/02. <u>https://doi.org/10.1093/hmg/ddu733</u> PMID: <u>25552646</u>; PubMed Central PMCID: PMCPMC4375422.
- 77. Fromer M, Pocklington AJ, Kavanagh DH, Williams HJ, Dwyer S, Gormley P, et al. De novo mutations in schizophrenia implicate synaptic networks. Nature. 2014; 506(7487):179–84. Epub 2014/01/28. <u>https://doi.org/10.1038/nature12929</u> PMID: 24463507; PubMed Central PMCID: PMCPMC4237002.
- Maere S, Heymans K, Kuiper M. BiNGO: a Cytoscape plugin to assess overrepresentation of gene ontology categories in biological networks. Bioinformatics. 2005; 21(16):3448–9. Epub 2005/06/24. https://doi.org/10.1093/bioinformatics/bti551 PMID: 15972284.
- 79. Cline MS, Smoot M, Cerami E, Kuchinsky A, Landys N, Workman C, et al. Integration of biological networks and gene expression data using Cytoscape. Nat Protoc. 2007; 2(10):2366–82. Epub 2007/10/20. <u>https://doi.org/10.1038/nprot.2007.324</u> PMID: <u>17947979</u>; PubMed Central PMCID: PMCPMC3685583.
- Merico D, Isserlin R, Stueker O, Emili A, Bader GD. Enrichment map: a network-based method for gene-set enrichment visualization and interpretation. PLoS One. 2010; 5(11):e13984. Epub 2010/11/ 19. <u>https://doi.org/10.1371/journal.pone.0013984</u> PMID: <u>21085593</u>; PubMed Central PMCID: PMCPMC2981572.